Author's response to reviews

Title: Clinico-pathogenetic findings and management of chondrodystrophic myotonia (Schwartz-Jampel syndrome)

Authors:

Dr Nicola C. Ho (honi@grc.nia.nih.gov)
Stacey S. Sandusky (sanduskyst@grc.nia.nih.gov)
Victor Madike (madikevi@grc.nia.nih.gov)
Dr Clair A. Francomano (francomanoccl@grc.nia.nih.gov)
Marinos M Dalakas (dalakasm@ninds.nih.gov)

Version: 2 Date: 16 Jun 2003

Dear Editor
We have revised the manuscript taking into account of the reviewers’ advice.
(1) We have used myotonia instead of myotonica (title and text)
(2) We have replaced the brand name Tegretol with carbamazepine instead and we have also indicate the dosage prescribed.